Author's response to reviews

Title: In vivo measurement of tumor estradiol and Vascular Endothelial Growth Factor in breast cancer patients

Authors:

Stina Garvin (stina.garvin@lio.se)
Charlotta Dabrosin (chada@ibk.liu.se)

Version: 5 Date: 10 March 2008

Author's response to reviews:

RE: In vivo measurement of tumor estradiol and Vascular Endothelial Growth Factor in breast cancer patients
BMC Cancer MS# 1506450516156080
By Stina Garvin and Charlotta Dabrosin

1. The name of the ethical review board and reference number of the approval has been included in the material section page 5. We have also included that the study was performed in compliance with the Helsinki Declaration, page 5.

2. All references have been included one-by-one and do now appear as separate numbers in the text. Hopefully this will show the reverences correctly. I use a Mac with EndNote 10 so I cannot right click to get any information of the references. Please let me know if there still is a problem.